Author’s response to reviews

Title: Capture of Intraocular Lens Optic by Residual Capsular Opening in Secondary Implantation: Long-term follow-up

Authors:

Tian Tian (tiantianoph@163.com)
Chunli Chen (chenchunli125@163.com)
Haiying Jin (13916917206@163.com)
Lyu Jiao (cherryfcl@126.com)
Qi Zhang (13601972300@163.com)
Peiquan Zhao (zhaopeiquan@126.com)

Version: 3 Date: 23 Feb 2018

Author’s response to reviews:

1. The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files.

Reply: The statement of data has been changed to "The datasets used and/or analysed during the current study available from the corresponding author on reasonable request." (Line 220-222, Page 10).

2.2. Please state whether the diagram depicted in figure 2D is your own or taken from another source.

The diagram depicted in figure 2D is my own made by CoreldrawX6 software.